Author’s response to reviews

Title: The long-term effects of naprapathic manual therapy on back and neck pain - Results from a pragmatic randomized controlled trial

Authors:

Eva Skillgate (eva.skillgate@ki.se)
Tony Bohman (tony.bohman@ki.se)
Lena W Holm (lena.holm@ki.se)
Eva Vingård (eva.vingard@medsci.uu.se)
Lars Alfredsson (lars.alfredsson@ki.se)

Version: 5 Date: 26 January 2010

Author’s response to reviews:

Dear editor

We have made some minor changes in the output regarding the table format and also some minor changes in spelling etc. Nothing that changes the meaning of the text or the tables.

Sincerely
Tony Bohman and Eva Skillgate